Author’s response to reviews

Title: Association between the DNA methylations of POMC, MC4R, and HNF4A and metabolic profiles in the blood of children aged 7-9 years

Authors:

Eun Jin Kwon (friendkej1004@hanmail.net)
Young-Ah You (yerang02@naver.com)
Bohyun Park (bhppark@ewha.ac.kr)
Eun Hee Ha (eunheeha@ewha.ac.kr)
Hae Soon Kim (hyesk@ewha.ac.kr)
Hyesook Park (hpark@ewha.ac.kr)
Young Ju Kim (kkyj@ewha.ac.kr)

Version: 2 Date: 22 Mar 2018

Author’s response to reviews:

1) Conclusions Section

Currently the conclusions section is missing, please see the link below for details on what to include:

---›Thank you for point this out. We have added this information in revised manuscript accordingly.

2) Consent to Participate

Please include the details of consent to participate in the 'Ethical Approval and Consent to Participate' section

---› As the reviewer suggested, we have added the manuscript.

3) Author Information

Please remove the author information form the manuscript.
4) Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

----- We upload the clean version of revised manuscript and tables/figures.